Author’s response to reviews

Title: Acute illness associated with Ehrlichia sp. Panola Mountain from Atlanta, Georgia, USA: a case report

Authors:

Will K Reeves (will.reeves@ars.usda.gov)
Amanda D Loftis (adloftis@gmail.com)
William L Nicholson (wln4@cdc.gov)
Alan G Czarkowski (bkt3@cdc.gov)

Version: 5 Date: 12 March 2008

Author’s response to reviews:

Response to the requested formatting changes: All 4 of the minor formatting changes recommended (details pasted below) have been made.

Formatting changes requested

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city and country.

Please use sentence case for all headings, such as change Case Presentation to Case presentation (lower case p).

Consent - Please reposition the consent section between the acknowledgements and Authors’ contributions. There are no images accompanying the manuscript, please remove the text and any accompanying images.

References Cited - Please rename this section to References.